Author's response to reviews

Title: Cleidocranial dysplasia - A case report

Authors:

Nagarathna C (shaanrathna@gmail.com)
B.S. Shakuntala (shakubethur@yahoo.co.in)
Somy Mathew (drsonmy78@gmail.com)
Navin H.K. (navina30@yahoo.co.in)
Ratna Yumkham (ratna.yumkham@gmail.com)

Version: 4 Date: 6 February 2011

Author's response to reviews: see over
Cover Letter

Ref.: Submission of the manuscript to the BMC JMCR for publication entitled, “Cleidocranial dysplasia – A case report” by Dr. Nagarathna C, Dr. B.S. Shakuntala, Dr. Somy Mathew, Dr. Navin H.K and Dr. Ratna Y.

With regards, the manuscript entitled, “Cleidocranial dysplasia – A case report” is the original work of ours and has not been published or presented in part or whole elsewhere and is solely contributed to the JMCR. The authors hereby transfer, assign and otherwise convey all the copyright ownership to BMC JMCR, in the event that the manuscript is published by BMC JMCR. The authors have no financial conflicts or proprietary interests in the subject matter or materials discussed in the manuscript including employment, consultancies, stock ownership, honoraria and paid expert testimony. There are no non-financial conflicts, as well. Also, authors have no objections to reviewing or editing of the submission by the editor.

Thanking you.

Sincerely Yours,

Dr. Nagarathna C (Prof. & H.O.D)
Dr. B.S. Shakuntala (Prof.)
Dr. Somy Mathew (Assoc Prof.)
Dr. Navin H.K (Senior Lecturer)
Dr. Ratna Yumkham (PG Student)

Address for correspondence:

Dr. Nagarathna. C
Department of Pedodontics and Preventive Dentistry,
Rajarajeswari Dental College & Hospital
Mysore road, Bangalore - 560074,
Karnataka, INDIA
Ph: +91 9845581779
e-mail: shaanrathna@gmail.com
shakubethur@yahoo.co.in
drsmom78@gmail.com
navina30@yahoo.co.in
ratna.yumkham@gmail